Author’s response to reviews

Title: Access to Health care and Employment Status of People with disabilities in South India, the SIDE (South India Disability Evidence) Study

Authors:

Murthy Venkata S Gudlavalleti (murthy.gvs@iiphh.org)
Neena John (neena.sushil@iiphh.org)
Komal Allagh (komal.allagh@iiphh.org)
Jayanthi Sagar (jayanthi.sagar@iiphh.org)
Sureshkumar Kamalakannan (suresh.kumar@iiphh.org)
Srikrishna S Ramachandra (srikrishna.sr@iiphh.org)

Version: 5
Date: 13 October 2014

Author’s response to reviews:

Dear Editor,

Thank you for sending us the comments from the reviewer. We have now addressed the concerns of the reviewer. We hope that this response will satisfy the reviewer.

A detailed rejoinder to the comments is as follows:

Major Compulsory revisions:

1. This is not a ‘case-control’ study. In response to my earlier comment, the authors have removed “case control study from the title”. However the terms “control” and “case control design” still feature at several places in the text. These need to be removed.

Our apologies at overlooking this; we have now assiduously removed all such references and hope that this satisfies the reviewer.

2. Table 4: The p-values of the adjusted odds ratios need to be provided in table 4.

We have now incorporated the p-values in the table as suggested by the reviewer.

Regards,

Gudlavalleti Venkata S Murthy